PLOS ONE



RESEARCH ARTICLE

# Information on 'Overdiagnosis' in Breast Cancer Screening on Prominent United Kingdom- and Australia-Oriented Health Websites

# Alex Ghanouni<sup>1</sup>, Susanne F. Meisel<sup>1</sup>, Jolyn Hersch<sup>2</sup>, Jo Waller<sup>1</sup>, Jane Wardle<sup>1†</sup>, Cristina Renzi<sup>1</sup>\*

1 Health Behaviour Research Centre, Department of Epidemiology and Public Health, University College London, 1–19 Torrington Place, London, WC1E 6BT, United Kingdom, 2 School of Public Health, University of Sydney, Edward Ford Building, Sydney, NSW 2006, Australia

† Deceased. \* <u>c.renzi@ucl.ac.uk</u>

# Abstract

#### **Objectives**

Health-related websites are an important source of information for the public. Increasing public awareness of overdiagnosis and ductal carcinoma in situ (DCIS) in breast cancer screening may facilitate more informed decision-making. This study assessed the extent to which such information was included on prominent health websites oriented towards the general public, and evaluated how it was explained.

### Design

Cross-sectional study.

### Setting

Websites identified through Google searches in England (United Kingdom) and New South Wales (Australia) for "breast cancer screening" and further websites included based on our prior knowledge of relevant organisations.

### **Main Outcomes**

Content analysis was used to determine whether information on overdiagnosis or DCIS existed on each site, how the concepts were described, and what statistics were used to quantify overdiagnosis.

#### Results

After exclusions, ten UK websites and eight Australian websites were considered relevant and evaluated. They originated from charities, health service providers, government



# G OPEN ACCESS

**Citation:** Ghanouni A, Meisel SF, Hersch J, Waller J, Wardle J, Renzi C (2016) Information on 'Overdiagnosis' in Breast Cancer Screening on Prominent United Kingdom- and Australia-Oriented Health Websites. PLoS ONE 11(3): e0152279. doi:10.1371/journal.pone.0152279

Editor: Anna Sapino, University of Torino, ITALY

Received: October 29, 2015

Accepted: March 12, 2016

Published: March 24, 2016

**Copyright:** © 2016 Ghanouni et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data are available on publicly accessible websites, which are cited in the manuscript (References: [7, 17–33]): http://www. cancerresearchuk.org/; http://www. againstbreastcancer.org.uk/; http://www. breastcancercare.org.uk/; http://breastcancernow.org/ ; http://www.healthtalk.org/; http://breastcancernow.org/ ; http://www.healthtalk.org/; http://www.macmillan.org. uk/; http://www.nhs.uk/Pages/HomePage.aspx; https://www.gov.uk/government/organisations/publichealth-england/; http://patient.info/; https://www.gov. uk/government/publications/breast-screening-helpingwomen-decide/; https://www.bcna.org.au/; http://www. cancer.org.au/; http://www.cancercouncil.com.au/;



http://www.cancerscreening.gov.au/internet/ screening/publishing.nsf/Content/breast-screening-1/; https://www.breastscreen.nsw.gov.au/; http://www. aihw.gov.au/; https://canceraustralia.gov.au/; https:// www.cancerinstitute.org.au/.

**Funding:** The current study was supported by a programme grant from Cancer Research UK awarded to Prof Jane Wardle (C1418/A14134). Cancer Research UK was not involved in the design of this study; the collection, analysis, or interpretation of the results; in the writing of the manuscript; or in the decision to submit for publication.

**Competing Interests:** The authors have declared that no competing interests exist.

agencies, and an independent health organisation. Most contained some information on overdiagnosis (and/or DCIS). Descriptive information was similar across websites. Among UK websites, statistical information was often based on estimates from the Independent UK Panel on Breast Cancer Screening; the most commonly provided statistic was the ratio of breast cancer deaths prevented to overdiagnosed cases (1:3). A range of other statistics was included, such as the yearly number of overdiagnosed cases and the proportion of women screened who would be overdiagnosed. Information on DCIS and statistical information was less common on the Australian websites.

# Conclusions

Online information about overdiagnosis has become more widely available in 2015–16 compared with the limited accessibility indicated by older research. However, there may be scope to offer more information on DCIS and overdiagnosis statistics on Australian websites. Moreover, the variability in how estimates are presented across UK websites may be confusing for the general public.

# Introduction

Overdiagnosis can be defined as the detection of a disease that would not cause symptoms or death if it had remained undetected. It is a widely debated issue in breast cancer screening since it can lead to unnecessary treatment (i.e. 'overtreatment'), including surgery, radiotherapy, hormone therapy, and chemotherapy. These can be associated with both short- and long-term adverse effects in addition to the negative psychological consequences of becoming a breast cancer patient [1]. Much overtreatment in breast screening is suspected to arise from the unnecessary and unbeneficial detection of ductal carcinoma in situ (DCIS). Although DCIS can become invasive, symptomatic, and a threat to life, it can also remain confined to the milk ducts in which it arose, remaining asymptomatic throughout a woman's lifetime [2].

Previous research has found both limited public awareness of overdiagnosis and widespread misconceptions about it. When a survey sample of 500 Australian adults were asked what they thought the term meant, only 41% were able to provide a definition that was considered approximately correct (e.g. *"exaggerating something that's there"*) [3] and a similar United Kingdom (UK) study of 390 adults found that less than 1% of the general public could provide a more precise definition (e.g. *"diagnosis of a disease that will not cause symptoms or death"* (Ghanouni et al., under review). Members of the public often have difficulty understanding the concept, and find it counterintuitive when it is explained [4, 5]. This suggests that there is substantial scope to improve how information on overdiagnosis is communicated.

A 2012 US survey found that 54% of adults aged 50 to 64 years had used the internet to search for health information in the past year [6], suggesting that health websites are likely to be an important way in which health concepts and recommendations are disseminated. In the UK, the Cancer Research UK website [7] alone regularly receives over 1 million visits per month [8]. However, a previous study evaluating 27 websites on breast cancer screening concluded that information on overdiagnosis (and overtreatment) was often omitted or inaccurate [9]. To our knowledge, although this study was published in 2004 (websites themselves were accessed in 2002), it is the most recent appraisal of online information relating to overdiagnosis in breast screening. However, there have been notable developments since then. For example, controversy around whether screening invitees were being adequately informed about

overdiagnosis [10] resulted in the commissioning of the Independent UK Panel on Breast Cancer Screening review (IUKPBCS) [2, 11] to estimate the magnitude of overdiagnosis and breast cancer mortality reduction attributable to screening. Although health websites might be expected to have incorporated the review's findings into their information materials, this has not been formally evaluated. Moreover, the web in 2016 bears little resemblance to the web as it existed in 2002. Availability of broadband and internet usage among the general public have increased [12], creating a greater incentive for producers of health content to provide highquality web-based information [13]. In addition, the increased bandwidth available to home users has allowed content producers to offer information in formats other than simple text and images, such as streaming video.

We evaluated information on overdiagnosis and DCIS provided by health websites oriented towards the general public in the UK and Australia (and England and New South Wales). These two countries were selected to allow an international analysis of English-language websites in areas where breast cancer screening is offered as part of organised screening programmes that is free at the point of use. We looked at whether any such information was provided, its content, and how it compared across websites, with particular focus on quantitative information describing the magnitude of overdiagnosis in breast screening.

# **Materials and Methods**

# Website identification

Google has approximately a 90% market share among search engines [14], making it an important intermediary between individuals seeking information on breast cancer screening and the information source itself. Consequently, websites were selected based on a pair of Google searches for "breast cancer screening", carried out on 13<sup>th</sup> August 2015 for UK-oriented websites and 14<sup>th</sup> December 2015 for Australia-oriented websites. These were performed using an "incognito" window in Chrome (for Windows; Google, Mountain View, CA, USA) in order to prevent cookies and user account data from influencing search results. The two searches were carried out in the respective countries of interest, in order to ensure that results were tailored to users from those countries. The first five pages of results (using the default number of results per page) were scanned manually in order to identify potentially relevant websites. This was anticipated to identify the vast majority of websites that users might realistically access: "clickthrough rates" for links are less than 5% for websites listed lower than the top ten results [15].

The Google search was supplemented with the authors' prior knowledge of resources that contained relevant information and were anticipated to be commonly used. Search results were excluded based on the following criteria: i) duplicate information sources (e.g. two links to the same website), suggested secondary search queries (e.g. "breast cancer screening age"), or subsidiary links (e.g. search results linking to information pages on "diagnosis" and "treatment" within a single main result); ii) research articles, technical reports, or sites aimed at healthcare professionals; iii) media releases or media articles (e.g. news stories); iv) websites from organisations with a remit outside the UK or Australia, as applicable (e.g. the National Institutes of Health in the United States); v) websites with a remit specific to a country (or state/territory) within the United Kingdom (or Australia) other than England (or New South Wales; NSW); vi) locally- or regionally-oriented websites (e.g. those of city-specific or county-wide breast screening services). Although the web is generally accessible across national borders, healthcare and screening systems are organised at a national level within the UK and at a state level in Australia. Hence, we limited our analysis to websites from organisations that were either focused on breast screening in England or NSW, or that related to the UK or Australia as a whole, in order to obtain suitably homogenous sets of resources that could be interpreted

within similar policy contexts. Finally, we excluded websites that did not provide information on breast cancer screening.

#### Analysis

Websites were categorised as originating from either charities, health service providers, government agencies, or other groups. They were evaluated using content analysis [16] to determine i) whether they included any information on overdiagnosis, overtreatment or DCIS in breast screening, ii) how overdiagnosis or DCIS was described, and iii) how overdiagnosis was quantified. Three authors browsed through websites manually with the aim of identifying all primary information pages that referred to any of these aspects. Browsing was supplemented by using the search functions built into websites, where these were available. Where relevant information was located, the text was extracted and saved in an Excel 2010 spreadsheet (for Windows; Microsoft, Redmond, WA, USA) for subsequent interpretation and comparison. Descriptions of overdiagnosis, DCIS, and relevant statistics were evaluated independently by pairs of authors who read the extracted text to identify what they considered to be the key information points, which were then listed and summarised for each website. These were then counted and (where possible) combined, in order to generate the number of websites containing each information point. Authors compared their categorisations in case of any discrepancies, which were resolved through discussion. The content of information was also compared across websites to highlight similarities and differences.

Since all data were publically available and the study did not involve human participants, the host institution considered this study to be exempt from requiring ethical approval.

## Results

## Website selection

The initial Google searches yielded 167 and 105 links for the UK and Australia, respectively. Duplicate results, suggested secondary searches, and secondary subsidiary links were excluded (UK: n = 112; Australia: n = 57), leaving 55 and 48 unique search results.

UK websites: Nine websites consisting of research articles, technical reports, or resources aimed at healthcare professionals were excluded, as were 14 media articles, six websites with a remit outside the UK, five websites with a remit specific to a UK country other than England, and 11 regionally-focused websites within England. The ten remaining websites were searched for information relating to breast cancer screening and a further three were excluded for not providing any such information. Three websites were added based on authors' prior knowledge for a total of ten websites to be included in the analysis.

Australian websites: Five research articles/professional resources and nine media articles were excluded. Nine websites with a remit outside Australia, nine region-specific websites outside NSW, and three region-specific websites within NSW were also excluded. Of the remaining websites, five were excluded for not including information about breast screening, leaving a total of eight websites to be analysed.

#### Website characteristics

UK websites: Six websites were from health charities (Against Breast Cancer, Breast Cancer Care, Breast Cancer Now, Cancer Research UK, healthtalk.org, Macmillan Cancer Support) [7, 17–21], two were from health service providers (National Health Services [NHS] Choices, and Public Health England) [22, 23], and one was from an independent health website (Patient) [24]. In England, invitees to the NHS Breast Screening Programme are routinely posted an

invitation leaflet that aims to provide enough information to allow women to make an 'informed choice' about their participation. An electronic version of the leaflet is accessible to the general public online (currently available via Public Health England but hosted on a separate website at the time of the analysis). Consequently, we evaluated this as well [25].

Australian websites: Three websites were from cancer charities (Breast Cancer Network Australia, Cancer Council Australia and Cancer Council NSW) [26-28], two were from breast screening service providers (BreastScreen Australia and BreastScreen NSW) [29-30], and three were from government agencies (Australian Institute of Health and Welfare, Cancer Australia, and Cancer Institute NSW) [31-33]. There was some overlap between organisations: the Cancer Council NSW is a member of the Cancer Council Australia (the national body of the charity). BreastScreen NSW is part of BreastScreen Australia (the national breast cancer screening programme) and is managed by the Cancer Institute NSW. As with the English Breast Screening Programme, the standard information leaflet provided to invitees in NSW is freely available online. It is hosted on the BreastScreen NSW website and so we considered it part of that resource for the purposes of this study.

#### Overdiagnosis and DCIS information

**Descriptions within websites.** UK websites: All but one of the websites included at least some information on overdiagnosis. These nine websites used similar general definitions, referring to it in terms of detection of cancer that would never have caused a problem, harm, symptoms, or become life threatening if left undetected. Three websites used both the terms 'overdiagnosis' and 'overtreatment' (or terms with this same word root e.g. 'overdiagnose'), and a further two websites used one of the terms. Six websites mentioned that it was not possible to determine which breast cancers would become invasive, grow quickly, or become life threatening for a given person diagnosed with breast cancer through screening. Seven websites also raised the issue that (most) people diagnosed with breast cancer would receive treatment and that this may be unnecessary in some cases. Furthermore, two websites made a categorical statement regarding the harms of overtreatment in terms of emotional, psychological and physical side effects.

Nine out of the ten websites included some information on DCIS, describing it as a form of cancer that had not invaded the surrounding breast tissue (or spread beyond the milk ducts). Of these nine websites, eight stated that it could become an invasive breast cancer if left untreated and six related the disease to the concept of overdiagnosis. Eight websites also noted that DCIS and screening were closely related, and four made the explicit point that treatment for DCIS was highly effective or that the prognosis was very good. On seven websites, a link to overdiagnosis was made in the form of statements that it was impossible to determine which cases of DCIS would become invasive for a given person.

Australian websites: Six of the eight Australian websites included at least some information on overdiagnosis. Information provided was similar across websites: five described overdiagnosis in terms of detection of breast cancer that would never have caused symptoms, health problems, or become life threatening. Five websites used the term 'overdiagnosis' and one also used the term 'overtreatment'. Six websites stated that it was impossible (or difficult) to determine whether a specific breast cancer would become invasive or harmful. Two websites stated that most people diagnosed with breast cancer would receive treatment, and one stated that unnecessary treatment was associated with harms.

The kinds of information provided about DCIS were similar to the UK websites: four websites gave a description of what it was (e.g. in terms of being a form of breast cancer that is currently confined to the milk ducts and has not become invasive). Two stipulated that DCIS could become invasive if left untreated (or even with treatment), and went on to state that it was not possible to determine whether a particular DCIS case would become invasive and that screening was a common pathway by which women might be diagnosed. Two websites were explicit that DCIS was the type of breast cancer to which overdiagnosis was more likely to apply, and one stated that most women treated for DCIS would not go on to develop invasive breast cancer.

Statistics within websites. UK websites: Websites varied considerably in terms of the statistics used to describe the magnitude of overdiagnosis. There were a few recurring examples: the statement that 4,000 women (in the UK) are overdiagnosed or overtreated (through screening) each year was present on three websites. Two websites stated that 3 in 200 women screened (every three years between age 50 and 70 years) would be overdiagnosed or overtreated, and two websites expressed a similar proportion using a different numerator and denominator (17 in 1,000, 129 in 10,000). Two stated a different proportion, seemingly based on a different reference group (women screened over 10 years: 10 in 2,000 or 1 in 200). Six websites stated the ratio of deaths from breast cancer prevented to overdiagnosed cases. In most cases this ratio was explicit, but a small amount of arithmetic was required from the reader for one website. Five websites used estimates derived from the IUKPBCS report when they stated the ratio to be 1:3 or (in one case) 5:17. However, one website stated the ratio to be between 2:1 and 2.5:1 based on research from another group [34]. Furthermore, only three websites referred to the IUKPBCS report explicitly when describing the magnitude of overdiagnosis. The proportion of screen-detected breast cancers that were overdiagnosed or overtreated was stated to be 1 in 4 on one website but 1 in 5 on another. Notably, in seven instances relating to the proportion of women screened who were overdiagnosed, or the ratio of breast cancer deaths prevented to overdiagnosed cases, the statistics provided differed from those provided in the information leaflet for the English NHS Breast Screening Programme. Finally, two websites provided no statistics relating to overdiagnosis.

Australian websites: Despite providing similar conceptual information on overdiagnosis and DCIS, there was relatively little overlap in the kinds of statistical information provided on Australian and UK websites. Two Australian websites referred to World Health Organisation estimates that one or two overdiagnoses of breast cancers occur for each breast cancer death avoided and went on to state that the lifetime risk of overdiagnosis was 3% (although the denominator was not explicitly stated). One other website stated four unique pieces of statistical information, namely i) that previous estimates of the proportion of overdiagnosed breast cancers (out of all breast cancers) in the range of 5–13% had been superseded by more recent studies, ii) that one recent review reported a range of 0–35% (median: 5–9%), iii) that another review reported a range of 1–10%, and iv) the IUKPBCS reported estimates of overdiagnoses in the range of 11–19%. Finally, this website referred to the approximate number of overdiagnoses per 1,000 women screened biennially from age 50 to 74 (around 8; between 2 and 21). This statistic was also presented on one other website. No other statistical information was identified on the remaining four websites.

### Discussion

In this 2015–2016 assessment of the most prominent health websites providing information to the general public on breast cancer screening, we found that almost all UK resources contained information on overdiagnosis and described it using similar terms. In addition, most UK websites explained that overdiagnosis would entail (unnecessary) treatment. Similarly, most UK websites provided information on DCIS, its relevance to breast cancer screening, and either related it to overdiagnosis explicitly or explained that it was unknown whether it would become

invasive for a given person. The majority of UK websites also used some form of statistics in order to illustrate the magnitude of overdiagnosis, and it was clear that the IUKPBCS review had been influential. Although it was not always cited explicitly, the statistics most commonly provided (the proportion of women screened who would be overdiagnosed and the ratio of breast cancer deaths prevented to overdiagnosed breast cancers) were derived from the IUKPBCS and there were few exceptions. However, although statistics on overdiagnosis were consistent in many cases, the way that they were presented differed substantially. For example, the probability of overdiagnosis for women undergoing screening was described as a proportion out of 200, 1,000, or 10,000.

A majority of Australian websites also provided information related to overdiagnosis. However, explicit information that a breast cancer diagnosis could lead to unnecessary treatment was less common. Most Australian websites also included information about DCIS although this was more variable: two websites included most of our key information points whereas remaining websites included either minimal information or no information. The Australian website that contained the most information relating to DCIS was also the most detailed website in terms of providing statistical information, including four estimates of the proportion of breast cancers that were overdiagnoses; only three other websites provided statistical details.

This study was strengthened by the selection of websites from the first five pages of Google search results. This was anticipated to generate a wide-ranging list of the most commonly accessed websites, less frequently visited websites, and websites that most users would never see. However, since search engine algorithms adjust results based on variables beyond the specific search terms (e.g. users' locations), website listings are unlikely to have been exhaustive.

Our findings provide an updated assessment compared to a previous study that reviewed websites in 2002, in which only 26% of websites were found to contain any information on overdiagnosis and overtreatment, and only 37% referred to DCIS [9]. Notwithstanding possible differences in website exclusion criteria between studies, this represents an appreciable change over the past 14 years, most likely reflecting a greater emphasis placed on the issue (and perhaps informed decision-making in general) among those who are professionally involved with screening.

Although online information on overdiagnosis has become more widely available, the relatively limited information about DCIS and statistics on Australian websites may be a concern; to the extent that one considers it important to include this information, it may indicate scope for improvement. However, this finding should be interpreted in the context of the fact that several of the Australian organisations whose websites we included are related to one another, meaning that some websites provided links to more detailed information contained elsewhere.

We also observed variation in how relevant statistical information was presented, especially between UK websites. Given that public understanding of health statistics is limited [35], this may be confusing to readers of multiple websites. In the context of the currently limited understanding of the concept of overdiagnosis among the general public (Ghanouni et al., under review) [3–5], these findings suggest that more could be done to provide clearer and possibly more consistent quantitative information on websites that describe breast cancer screening. This is particularly important where statistics differ from those provided by healthcare providers themselves (e.g. the English NHS Breast Screening Programme and BreastScreen NSW), as these are the set most likely to be encountered by women invited to screening. However, as the most detailed Australian website notes, various estimates of the magnitude of overdiagnosis exist and it is a matter of intense debate as to which are most accurate. Hence, an aim of providing consistent information would need to be balanced against another defensible goal of informing women that there is uncertainty and that several sets of statistics exist.

These findings suggest several possibilities for further research. The analysis consisted primarily of websites that contained content produced by either a health charity or healthcare provider. However, an appreciable amount of public exposure to information on breast cancer screening online is likely to occur on websites without a specific health focus (e.g. social media, news, or video hosting and streaming services) [36]. It may be useful for future research to explore the nature of overdiagnosis information on websites such as Twitter, YouTube, and those of mainstream media outlets. Furthermore, in order to limit the role of subjective personal opinions in our evaluations, we limited our information coding to broad descriptive categories. Future studies could also examine other aspects relating to how that content is communicated, such as information placement (e.g. if information on overdiagnosis was less discoverable due to being located at the bottom of a lengthy page), tone (e.g. whether information was empathetic or emotionally detached), and the use of visual elements to describe statistics (e.g. graphs or icon arrays).

In conclusion, we found that a higher proportion of UK and Australian health websites explained overdiagnosis and DCIS than websites accessed in 2002, although there was scope for improvement within both sets. Descriptive information on the concepts was relatively consistent, although information relating to DCIS and the relevant statistics were less prevalent among Australian websites. Statistical information varied considerably across UK websites, potentially causing confusion. Future work could extend the area of study to websites that are not health-specific and describe more subtle characteristics of website information.

# Acknowledgments

The authors thank Jessica Haddrell and Athena Ip for their assistance with an initial scoping exercise and preliminary analysis of health-related websites.

# **Author Contributions**

Conceived and designed the experiments: AG SFM JH J. Waller J. Wardle CR. Performed the experiments: AG SFM JH. Analyzed the data: AG SFM JH. Contributed reagents/materials/ analysis tools: AG SFM JH. Wrote the paper: AG SFM JH J. Waller J. Wardle CR.

# References

- 1. Glasziou P, Moynihan R, Richards T. Too much medicine; too little care. BMJ 2013; 347:f4247. doi: <u>10.</u> <u>1136/bmj.f4247</u> PMID: <u>23820022</u>
- Independent UK Panel on Breast Cancer Screening. The benefits and harms of breast cancer screening: an independent review. Lancet 2012; 380:1778–86. doi: <u>10.1016/S0140-6736(12)61611-0</u> PMID: <u>23117178</u>
- Moynihan R, Nickel B, Hersch J, Doust J, Barratt A, Beller E, et al. What do you think overdiagnosis means? A qualitative analysis of responses from a national community survey of Australians. BMJ Open 2015; 5:e007436. doi: <u>10.1136/bmjopen-2014-007436</u> PMID: <u>25991454</u>
- Hersch J, Jansen J, Barratt A, Irwig L, Houssami N, Howard K, et al. Women's views on overdiagnosis in breast cancer screening: a qualitative study. BMJ 2013; 346:f158. doi: <u>10.1136/bmj.f158</u> PMID: <u>23344309</u>
- Waller J, Douglas E, Whitaker KL, Wardle J. Women's responses to information about overdiagnosis in the UK breast cancer screening programme: a qualitative study. BMJ Open 2013; 3:e002703. doi: <u>10.</u> <u>1136/bmjopen-2013-002703</u> PMID: <u>23610383</u>
- Pew Research Center. Health Fact Sheet 2015 [cited 2015 Oct 13]. Available from: <u>http://www.pewinternet.org/fact-sheets/health-fact-sheet/</u>.
- 7. Cancer Research UK [cited 2015 Oct 13]. Available from: http://www.cancerresearchuk.org/.
- Cancer Research UK. Cancer Research UK's strategy 2009–2014, 2008. Available from: <u>http://www.cancerresearchuk.org/prod\_consump/groups/cr\_common/@abt/@gen/documents/generalcontent/cr\_043314.pdf</u>.

- 9. Jørgensen KJ, Gøtzsche PC. Presentation on websites of possible benefits and harms from screening for breast cancer: cross sectional study. BMJ 2004; 328:148. PMID: <u>14726344</u>
- Bewley S. The NHS breast screening programme needs independent review. BMJ 2011; 343:d6894. doi: <u>10.1136/bmj.d6894</u> PMID: <u>22028484</u>
- 11. Richards M. An independent review is under way. BMJ 2011; 343:d6843. doi: <u>10.1136/bmj.d6843</u> PMID: <u>22028483</u>
- 12. OFCOM. Household penetration of broadband [cited 2015 Oct 13]. Available from: <a href="http://stakeholders.ofcom.org.uk/market-data-research/market-data/communications-market-reports/cmr11/uk/1.3">http://stakeholders.ofcom.org.uk/market-data-research/market-data/communications-market-reports/cmr11/uk/1.3</a>.
- Geller BM, Zapka J, Hofvind SS, Scharpantgen A, Giordano L, Ohuchi N, et al. Communicating with women about mammography. J Cancer Educ 2007; 22:25–31. PMID: <u>17570805</u>
- Statistica. Market share held by the leading search engines in the United Kingdom (UK) as of August 2015 [cited 2015 Oct 13]. Available from: <u>http://www.statista.com/statistics/280269/market-share-heldby-search-engines-in-the-united-kingdom/</u>.
- 15. Petrescu P, Ghita M, Loiz D. Google organic CTR study. 2014. Available from: <u>http://www.advancedwebranking.com/google-ctr-study-2014.html</u>.
- Fereday J, Muir-Cochrane E. Demonstrating rigor using thematic analysis: A hybrid approach of inductive and deductive coding and theme development. Int J Qual Methods 2006; 5:80–92.
- 17. Against Breast Cancer [cited 2015 Oct 13]. Available from: http://www.againstbreastcancer.org.uk/.
- 18. Breast Cancer Care 2015 [cited 2015 Oct 13]. Available from: https://www.breastcancercare.org.uk/.
- 19. Breast Cancer Now 2015 [cited 2015 Oct 13]. Available from: http://breastcancernow.org/.
- 20. Healthtalk.org 2015 [cited 2015 Oct 13]. Available from: http://www.healthtalk.org/.
- 21. Macmillan Cancer Support 2015 [cited 2015 Oct 13]. Available from: http://www.macmillan.org.uk/.
- 22. NHS Choices [cited 2015 Oct 13]. Available from: http://www.nhs.uk/Pages/HomePage.aspx.
- Public Health England [cited 2015 Oct 13]. Available from: <u>https://www.gov.uk/government/organisations/public-health-england</u>.
- 24. Patient [cited 2015 Oct 13]. Available from: http://patient.info/.
- NHS Breast Screening Programme. NHS Breast Screening: Helping you decide [cited 2016 Feb 26]. Available from: <u>https://www.gov.uk/government/publications/breast-screening-helping-women-decide</u>.
- 26. Breast Cancer Network Australia 2016 [cited 2016 Mar 01]. Available from: https://www.bcna.org.au/.
- 27. Cancer Council Australia [cited 2016 Mar 01]. Available from: http://www.cancer.org.au/.
- 28. Cancer Council NSW 2015 [cited 2016 Mar 01]. Available from: http://www.cancercouncil.com.au/.
- BreastScreen Australia [cited 2016 Mar 01]. Available from: <u>http://www.cancerscreening.gov.au/</u> internet/screening/publishing.nsf/Content/breast-screening-1.
- 30. BreastScreen NSW 2016 [cited 2016 Mar 01]. Available from: https://www.breastscreen.nsw.gov.au/.
- **31.** Australian Institute of Health and Welfare 2016 [cited 2016 Mar 01]. Available from: <u>http://www.aihw.gov.au/</u>.
- 32. Cancer Australia 2016 [cited 2016 Mar 01]. Available from: https://canceraustralia.gov.au/.
- 33. Cancer Institute NSW 2016 [cited 2016 Mar 01]. Available from: https://www.cancerinstitute.org.au/.
- 34. Duffy SW, Tabar L, Olsen AH, Vitak B, Allgood PC, Chen TH, et al. Absolute numbers of lives saved and overdiagnosis in breast cancer screening, from a randomized trial and from the Breast Screening Programme in England. J Med Screen 2010; 17:25–30. doi: <u>10.1258/jms.2009.009094</u> PMID: <u>20356942</u>
- Lipkus IM, Samsa G, Rimer BK. General performance on a numeracy scale among highly educated samples. Med Decis Making 2001; 21:37–44. PMID: <u>11206945</u>
- Basch CH, Hillyer GC, MacDonald ZL, Reeves R, Basch CE. Characteristics of YouTube<sup>™</sup> Videos Related to Mammography. J Cancer Educ 2014; 30:699–703.